Author's response to reviews

Title: Definition, etiology, prevention and treatment of peri-implantitis - a review

Authors:

Ralf Smeets (r.smeets@uke.de)
Anders Henningsen (a.henningsen@uke.de)
Ole T Jung (ole.tiberius.jung@gmail.com)
Max P Heiland (m.heiland@uke.de)
Christian Hammächer (chammaecher@arcor.de)
Jamal M Stein (jstein@ukaachen.de)

Version: 2
Date: 20 July 2014

Author’s response to reviews:

Editor-in-Chief
Journal of Head and Face Medicine
PD. Dr. Thomas Stamm
19nd of july, 2014

Dear Dr. Stamm,

enclosed please find our manuscript entitled “Definition, etiology, prevention and treatment of peri-implantitis – a review” by Ralf Smeets, Anders Henningsen, Ole Jung, Max Heiland, Christian Hammächer, Jamal Stein.

We carefully read the reviewers’ comments and recommendations and unfortunately have to agree to the criticism. In closing we would like to express our appreciation for all the value inputs which will greatly improve the quality of this paper. All changes done in the manuscript are marked red.

Sincerely yours,

Ralf Smeets, M.D., D.M.D., PhD
Associate Professor of Oral and Maxillofacial Surgery